
Stacking
○ Stacked 

� <stacking rate> = 25.1 ma/hr
� <production> =  21.6e-6/proton
� <beam on target> = 7.71 at 11 turns, 8.25 at 12 turns

○ Pbar Crate $33 Slot #1 Camac 190 card (connected to MADC #16) had a No-Q and 
reset ok.   This C190 has a history of problems

� In April 23 the C190 card was replaced
� On April 27 the MADC was replaced.
� Controls experts informed us that they also have changed out the cabling 

between the MADC and C190.
○ Ops did a good job of performing the normal tuning efforts over the weekend.

Transfers
○ Unstacked 1623e10 in 155 transfers over 71 sets

� Overall efficiency from Accumulator to Recycler was approximately 96%
� Taking out the sets with more than 2 transfers, we 96.5%.

2009-05-04 Monday Pbar Notes Friday, May 01, 2009
1:45 PM
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Studies

Requests

The Numbers
○ Pauls numbers
○ Most in a half hour: 13.40 mA at Mon May 04 03:57:55 CDT 2009 
○ Best Hour: 28.56 mA on 20-Dec-08 
○ Average Production 19.05 e-6/proton Best: 25.41 e-6/proton on 01/30/2008 
○ Average Protons on Target 7.98 e12 Best: 8.77 e12 on 07/24/2007 

○ Als numbers

� Stacking
□ Pbars stacked: 1620.27 E10
□ Time stacking: 67.42 Hr
□ Average stacking rate: 24.03 E10/Hr

� Uptime
□ Number of pulses while in stacking mode: 102798
□ Number of pulses with beam: 98868
□ Fraction of up pulses was: 96.18%

� The uptime's effect on the stacking numbers
□ Corrected time stacking: 64.84 Hr
□ Possible average stacking rate: 24.99 E10/Hr
□ Could have stacked: 1684.67 E10/Hr

� Recycler Transfers
□ Pbars sent to the Recycler: 1622.93 E10
□ Number of transfers : 155
□ Number of transfer sets: 71
□ Average Number of transfer per set: 2.18
□ Time taken to shoot including reverse proton tuneup: 00.95 Hr
□ Transfer efficiency: 96.06%
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□ Transfer efficiency: 96.06%

� Other Info
□ Average POT : 8.04 E12
□ Average production: 20.39 pbars/E6 protons

� * Missed one or more A:IBEAM7 events somewhere in the middle of the user 
selected time span. Calculated time shot using 13 secs per transfer.

Misc
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Logbooks

○ MCR Elog:

� Friday, May 1
□ 12:02:45-  R:KPS4A tripped with a "Filament Ready" and a "Tube Temp" 

fault. We are contacting experts. 
Pasted from <http://www-bd.fnal.gov/cgi-mcr/elog.pl?nb=2009&action=view&page=last&frame=2
&scroll=true> 

♦ Fri May 1 13:47:34 comment by...prk --  The "Tube Temp" bit 
has cleared but the filament is not resetting. Dirong Chen is 
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has cleared but the filament is not resetting. Dirong Chen is 
investigating. 

♦ Fri May 1 14:04:24 comment by...m.m --  Dirong made some 
adjustments to the resivoir voltage and was able to get the 
module reset. 

� Saturday
□ Pbar Crate 33, Slot 1 MADC #16 - No Q errors

♦ Have reset multiple times.

□

○ Pbar Elog:
○ RunCo Elog:
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○
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